```
Mauna Loa Solar Observatory Observer's Log
 ______
       Wed Dec 6 16:59:37 GMT 2006
Year: 06 Doy: 340
Observer: yasukawa
WEATHER COMMENT: Wed Dec 6 16:59:58 GMT 2006
In thin cirrostratus, sky very bright near sun, S wind 11 mph, temp 41 F.
  end
                                      Start Patrol
Wed Dec 6 17:12:23 GMT 2006
                              CHIP
Wed Dec 6 17:12:33 GMT 2006
                            PICS
                                      Start Patrol
**SBM PROBLEM**: Wed Dec 6 17:14:24 GMT 2006
Communication with camera was broken. Exited sbm program, cycled
instrument power, restarted.
end
SBM COMMENT: Wed Dec 6 17:15:21 GMT 2006
Observing.
end
A NEW TAPE HAS BEEN PUT INTO KAIEE DLT DRIVE, Wed Dec 6 17:15:53 GMT 2006
PSPT COMMENT: Wed Dec 6 17:17:09 GMT 2006
Starting up.
 end
PSPT COMMENT: Wed Dec 6 17:31:16 GMT 2006
Observing.
end
Wed Dec 6 18:00:56 GMT 2006
                              CHIP
                                      LSD
Wed Dec 6 18:01:13 GMT 2006
                              PICS
                                      Flat
Wed Dec 6 18:02:42 GMT 2006
                             CHIP
                                      End LSD
Wed Dec 6 18:02:48 GMT 2006
                                      BiasLSD
                             CHIP
Wed Dec 6 18:03:29 GMT 2006
                                      End Flat
                             PICS
Wed Dec 6 18:03:35 GMT 2006
                             PICS
                                      ReStart Patrol
Wed Dec 6 18:03:39 GMT 2006
                             CHIP
                                      End BiasLSD
Wed Dec 6 18:03:45 GMT 2006
                             CHIP
                                      Bias
Wed Dec 6 18:04:36 GMT 2006
                              CHIP
                                      End Bias
Wed Dec 6 18:04:41 GMT 2006
                             CHIP
                                      ReStart Patrol
**PSPT PROBLEM**: Wed Dec 6 18:26:59 GMT 2006
email indicated PSPT crashed but it is still running. Perhaps the
data transfer is not occurring. Aborting and restarting.
**** EVENT COMMENT ****: Wed Dec 6 18:49:41 GMT 2006
Large flare near east limb started at around 1830 UT.
 end
Wed Dec 6 19:09:30 GMT 2006
                             MKIV
                                     Start Patrol
Wed Dec 6 19:15:45 GMT 2006
                             MKIV End Patrol
NICE PICSDISC IMAGE: 1845
NICE CHIP
            IMAGE: 1847
NICE CHIP
            IMAGE: 1847
```

\*\*PSPT PROBLEM\*\*: Wed Dec 6 21:36:11 GMT 2006

```
Pspt crashed at 2122 UT, k/r.

__end__
COMMENT: Thu Dec 7 01:42:41 GMT 2006

Declination recentered itself. Repointed and reset guider.

__end__
**PSPT PROBLEM**: Thu Dec 7 02:26:29 GMT 2006

Crashed at 0218 UT, killed, shut down.

__end__
Thu Dec 7 02:28:48 GMT 2006 PICS End Patrol
Thu Dec 7 02:28:49 GMT 2006 CHIP End Patrol
Thu Dec 7 02:33:13 GMT 2006
```

MkIV